



# Article Sazgar IoT: A Device-Centric IoT Framework and Approximation Technique for Efficient and Scalable IoT Data Processing

Ali Yavari <sup>1,2,\*</sup>, Harindu Korala <sup>3</sup>, Dimitrios Georgakopoulos <sup>2</sup>, Jonathan Kua <sup>4</sup>, and Hamid Bagha <sup>2</sup>

- <sup>1</sup> 6G Research and Innovation Lab, Swinburne University of Technology, Melbourne, VIC 3122, Australia
- <sup>2</sup> School of Science, Computing and Engineering Technologies, Swinburne University of Technology,
- Melbourne, VIC 3122, Australia; dgeorgakopoulos@swin.edu.au (D.G.); hbagha@swin.edu.au (H.B.)
  <sup>3</sup> Institute of Railway Technology, Monash University, Melbourne, VIC 3800, Australia; hkorala@swin.edu.au
- <sup>4</sup> School of Information Technology, Deakin University, Geelong, VIC 3220, Australia;
  - jonathan.kua@deakin.edu.au
- \* Correspondence: mail@aliyavari.com

Abstract: The Internet of Things (IoT) plays a fundamental role in monitoring applications; however, existing approaches relying on cloud and edge-based IoT data analysis encounter issues such as network delays and high costs, which can adversely impact time-sensitive applications. To address these challenges, this paper proposes an IoT framework called Sazgar IoT. Unlike existing solutions, Sazgar IoT leverages only IoT devices and IoT data analysis approximation techniques to meet the time-bounds of time-sensitive IoT applications. In this framework, the computing resources onboard the IoT devices are utilised to process the data analysis tasks of each time-sensitive IoT application. This eliminates the network delays associated with transferring large volumes of high-velocity IoT data to cloud or edge computers. To ensure that each task meets its application-specific time-bound and accuracy requirements, we employ approximation techniques for the data analysis tasks of time-sensitive IoT applications. These techniques take into account the available computing resources and optimise the processing accordingly. To evaluate the effectiveness of Sazgar IoT, experimental validation has been conducted. The results demonstrate that the framework successfully meets the time-bound and accuracy requirements of the COVID-19 citizen compliance monitoring application by effectively utilising the available IoT devices. The experimental validation further confirms that Sazgar IoT is an efficient and scalable solution for IoT data processing, addressing existing network delay issues for time-sensitive applications and significantly reducing the cost related to cloud and edge computing devices procurement, deployment, and maintenance.

**Keywords:** Internet of Things; approximation; time-sensitive computing; contextualisation; COVID-19; pandemic

# 1. Introduction

The Internet of Things (IoT) is the latest Internet evolution that incorporates billions of sensors and other machines (we refer to these as IoT devices or Things) that support the development of IoT-based services and related smart products [1]. Currently, IoT devices (e.g., cameras, smartphones, sensors, smart vehicles, industrial machines, etc.) sense the physical world and send observation data (referred to as IoT data) to IoT-based software services that run in the cloud and edge computers. These computing resources and IoT devices are interconnected via a variety of networks ranging from broadband to low-power IoT networks such as Narrowband IoT (NB-IoT) and LoRa Wide Area Network (LoRaWAN). The unprecedented ability of the IoT ecosystem to observe the physical world and provide valuable information allows IoT-based services to address a magnitude of grand challenges in our society that were unsolvable before due to a lack of critical information from the physical world.



Citation: Yavari, A.; Korala, H.; Georgakopoulos, D.; Kua, J.; Bagha, H. Sazgar IoT: A Device-Centric IoT Framework and Approximation Technique for Efficient and Scalable IoT Data Processing. *Sensors* **2023**, *23*, 5211. https://doi.org/10.3390/ s23115211

Academic Editors: Paolo Bellavista and Jose Manuel Molina López

Received: 16 April 2023 Revised: 15 May 2023 Accepted: 29 May 2023 Published: 30 May 2023



**Copyright:** © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). Nevertheless, the full potential of the IoT ecosystem is still far from being fully realised because of the lack of comprehensive solutions that ensure the security and quality of IoT data, improve the reuse of existing IoT devices, provide better support for time-sensitive IoT (TS-IoT) applications, and better support critical industrial applications and industry 4.0 vision.

This paper addresses the challenge of TS-IoT applications (i.e., applications that require their processing to be completed within a specific time-bound, otherwise the produced results will not be useful for the application) [2,3]. TS-IoT applications are relatively common, and they range from preventing traffic accidents and reducing fuel theft in gas stations by preventing vehicles with stolen license plates to refuel to achieving a higher degree of automation control in manufacturing plants, monitoring and managing greenhouse gases, and so forth [4–7]. In this paper, we focus on a global issue that was critical in mitigating COVID-19 infections: How to use RGB/Infrared/Depth cameras and other sensors deployed in public places to detect citizen COVID-19 lockdown violations and related symptoms and how to do this at a minimal cost?

While it is currently possible to develop an IoT application that utilises IoT devices, such as cameras and sensors to monitor public places, existing IoT approaches for development such as the COVID-19 solution currently follow a 'device-to-cloud' architecture that involves IoT devices generating IoT data that are harvested and analysed by cloud-based IoT applications. A later evolution of this architecture includes using 'nearby' computers, such as edge computers, located between the IoT devices and the cloud to analyse the IoT device data streams [8–10].

Cloud-based data analysis is not ideal for time-sensitive applications because of the significant communication delays involves in sending large and high-velocity data streams for processing in a remote cloud data centre [2]. Edge computing has been introduced to reduce such network delays and related overhead. Unlike traditional cloud-based IoT data analysis, edge computing introduces computers that are located closer to the IoT devices that are used by the IoT applications at hand (the 'closeness' is determined by the network delay between the edge and IoT devices). Therefore, the latest trend at this point is focused on edge computing to distribute IoT applications so they can utilise a combination of the available cloud and edge computing resources. Such an approach makes it possible to compute time-sensitive data analysis tasks at the edge, which is 'closer' to the IoT devices involved in each data analysis task [8]. However, data processing at the edge is complex, subject to communication delays, and also introduces additional deployment and maintenance costs for edge computing (in addition to the IoT devices and cloud servers). Furthermore, deploying and maintaining (or utilising alternative third-party-provided) edge computers and cloud data centres is expensive, and in some cases, the costs of TS-IoT applications outweigh their benefits.

This paper focuses on developing a device-centric IoT framework, referred to as Sazgar IoT (Sazgar is a Persian word meaning adaptable), that can harvest the data and computing power of IoT devices in monitoring citizen compliance with COVID-19 government guidelines and restrictions as shown in Figure 1, e.g., counting people in public transport and other public places such as food stores, detecting and counting the ratio of people that wear face masks, and measuring social distancing. Such a complex application includes many time-sensitive data analysis tasks as compliance metrics must be computed before the situation changes (e.g., people come on board or closer, masks are taken off, etc.). Failure to do so will produce inaccurate information in real-time as public transport occupancy, social distancing, and mask-wearing can change rapidly.



Figure 1. Monitoring government guidance and restriction compliance.

Figure 1 outlines the time-sensitive monitoring of COVID-19 lockdown compliance and symptoms application, including its main tasks and the underlying physical infrastructure and the IoT devices used. The application consists of two main components.

The enforcing compliance component is responsible for monitoring how well COVID-19related compliance is adhered to by the public during the lockdowns in public areas. To realise this, the application estimates the social distance between people, detects unauthorised movement during curfew, monitors sanitiser usage, detects people who are wearing face masks, and computes occupancy in public places.

The other component of the application is the COVID-19 symptoms monitoring. This component monitors symptoms such as high fever and dry cough of people in public places using various IoT devices and determines potential suspicious COVID-19 patients in real-time. This information is reported to health officials in real time so that the necessary actions can be taken to delay the spread of the virus.

In Section 3, we will further discuss the application.

While video analysis algorithms and related Machine Learning (ML)-based models for identifying objects such as masks, counting people, and measuring their distance are well investigated [11–15], the main challenge in devising this important COVID-19 citizen compliance monitoring application is where and how to compute the related IoT data analysis task in a way that we can meet the application time-bounds so that the compliance measures computed by the COVID-19 citizen compliance monitoring application are accurate.

For example, the COVID-19 citizen compliance monitoring application must compute the number of passengers on a bus or train before the next bus stop, and the number of people in a supermarket by the time any person enters or leaves the supermarket. If this and other time-sensitive applications do not produce information within such application-specific time-bounds, such information is either not useful or problematic, i.e., the COVID-19 citizen compliance monitoring application will conclude that compliance has been achieved while this is not the case, or vice versa.

To address the above challenges, Sazgar IoT considers all related trade-offs, including the resource cost (i.e., purchasing, installing, servicing, and periodically replacing edge computers), the data analysis response time (i.e., the time-sensitive requirements in IoT applications) and the computing resource constraints of IoT devices.

Sazgar IoT meets the time-bounds of time-sensitive IoT applications by (1) analysing all IoT data that are consumed by a time-sensitive IoT application on the IoT devices that

generate them; (2) leveraging IoT application-specific time-bounds and descriptions of the computing resources onboard the IoT devices to approximate the IoT data and related analysis task(s) at each device in a way that such approximations of IoT data and/or analysis tasks achieves meeting the time-bounds of each IoT application and effectively manages the utilisation of the onboard computing resources of the devices; and finally, (3) eliminating the costs needed to purchase, deploy, use, and maintain edge computers and cloud data centres.

Existing IoT solutions as shown in Table 1 [8–10,16–22] explore how to best meet the IoT application requirements including Quality of Service (QoS), energy consumption, scalability, latency, security, interoperability, reliability, and cost by utilising the computing resources that are provided by cloud data centres, edge computers and/or IoT devices. However, the problem of meeting IoT application time-bounds by optimising the distribution of TS-IoT applications in such computing resources is NP-hard. In addition, existing research outcomes cannot deal with IoT device and data rate volatility.

Table 1. Comparison of the state-of-the-art in IoT solutions meeting the IoT application requirements.

	QoS	Energy Consumption	Scalability	Latency	Security	Interoperability	Reliability	Cost
[8]	$\checkmark$	$\checkmark$		$\checkmark$				$\checkmark$
[16]	$\checkmark$			$\checkmark$				$\checkmark$
[17]			$\checkmark$	$\checkmark$			$\checkmark$	
[18]		$\checkmark$		$\checkmark$	$\checkmark$			$\checkmark$
[9]	$\checkmark$		$\checkmark$	$\checkmark$	$\checkmark$			$\checkmark$
[19]							$\checkmark$	$\checkmark$
[10]	$\checkmark$			$\checkmark$		$\checkmark$	$\checkmark$	$\checkmark$
[20]	$\checkmark$			$\checkmark$				$\checkmark$
[21]			$\checkmark$				$\checkmark$	
[22]	$\checkmark$	$\checkmark$		$\checkmark$			$\checkmark$	$\checkmark$

This paper explores a device-centric approach that introduces approximation techniques for meeting the time-bounds of TS-IoT applications. Unlike other existing research, this approach does not involve solving NP-hard challenges and can deal with IoT device and data rate volatility.

More specifically, this paper includes the following contributions:

- An IoT device-centric approach that involves using the computing resources onboard the IoT devices to process the data analysis tasks of each TS-IoT application. This eliminates network delays for transferring large volumes of high-velocity IoT data (e.g., the RGB/Infrared/depth data streams in the COVID-19 citizen compliance monitoring) to the cloud and/or edge computers.
- 2. Approximation techniques for the data analysis tasks of TS-IoT applications that ensure each task meets its application-specific time-bound and accuracy requirements and at the same time its processing can be completed with the available computing resources onboard the IoT device that harvests the relevant IoT data. IoT task approximation eliminates the complexity of utilising cloud resources and edge computers as well as related cost and time frames for edge computer deployment and cloud resource lease (which are grant challenges in the COVID-19 citizen compliance monitoring).
- 3. A framework, which we refer to as Sazgar IoT, that combines and implements the above IoT device-centric and approximation approaches. Sazgar IoT can meet the requirements of COVID-19 citizen compliance monitoring (as well as other time-sensitive applications that are outside the scope of this paper).
- 4. An experimental validation of Sazgar IoT that shows the framework is able to address both time-bounds and accuracy requirements of the COVID-19 citizen compliance monitoring application by utilising only available IoT devices.

The remainder of the paper is organised as follows. Section 2 presents related work. Section 3 explains our COVID-19 citizen compliance application. The device-centric approach in processing TS-IoT applications using approximation is discussed in Section 4. Sazgar IoT framework and the functionality of each engine within the framework are presented in Section 5. Section 6 describes the experimental evaluation for this paper as well as the data analysis and results. Section 7 concludes the paper and outlines the potential future work.

# 2. Related Work

A large amount of work is rapidly building in the area of leveraging technological innovations in digital technologies to combat the COVID-19 pandemic. In this section, we review some of the related work in this area, primarily covering literature and surveys in the areas of using Artificial Intelligence (AI), the Internet of Things (IoT), Unmanned Aerial Vehicles (UAV), Blockchain, and 5G networks in the context of COVID-19. Innovations in these areas have transformed many different applications. The role of these technologies in fighting against COVID-19 is increasingly pronounced in different aspects, ranging from enforcing social distancing and mask compliance to deep analysis of medical imaging.

In a two-part paper, Nguyen et al. [23,24] provided a comprehensive survey of key enabling and emerging technologies for social distancing. The authors covered a wide range of systems and technologies and discussed how existing wireless and networking technologies and AI can enable, encourage, and enforce social distancing practices. In particular, they considered commonly-used communication technologies such as Wi-Fi, Cellular, Bluetooth, Ultra-wideband, ZigBee, and RFID in enforcing social distancing. Their work is followed by a discussion of the many open problems, solutions, and directions in dealing with issues surrounding the enforcement of social distancing, such as early symptom prediction, detection, and monitoring of people in quarantine, tracking people's movements, density prediction, and contact tracing. More advanced techniques using ML, computer vision, thermal, and ultrasound sensing systems are also discussed. Key issues such as security and privacy while handling personal sensitive data (such as location data) are presented for consideration. The paper also presents possible future responses to a global pandemic using smart infrastructures, smart cities, and intelligent transportation systems enabled by next-generation mobile technologies (5G and beyond).

Hussain et al. [25] surveyed the application of AI and big data in clinical administrations. The authors provided an overview of various techniques driven by AI and big data to be applied to medical information to help medical practitioners and researchers to better understand the virus. Neural systems, classical Support Vector Machine (SVM), and edge significant learning are techniques that are covered in their work. Their work investigated these techniques and demonstrated the effectiveness of using AI and big data to mine the large amount of information available, and also discussed how such techniques can be used in similar medical settings. In a similar vein, Pham et al. [26] also investigated the role of AI and big data in fighting against COVID-19. The authors surveyed a larger body of work, including using deep learning models for early detection and diagnosis; identifying, tracking, and predicting the outbreak; infodemiology and infoveillance; and biomedicine and pharmacotherapy.

In [27], Hossain et al. focused on investigating the use of tactile edge technology and hierarchical edge computing systems enabled by 5G to control and contain infectious diseases. By leveraging the advantages provided by the 5G network, such as its ultra-low latency, high bandwidth, and high reliability, the authors proposed a Beyond 5G (B5G) framework empowered by deep learning (DL) methods for fast processing of chest X-ray images and CT scans, and also enabled a mass surveillance system to monitor social distancing, mask-wearing, and body temperature using smart cameras. The proposed framework has three different in-built DL models, namely ResNet50, Deep tree, and Inception v3. Another aspect of this framework is the use of blockchain to ensure the security of patients' personal healthcare data. Fog data encryption has also been the focal point of Raghunandan et al. [28] where they used chaotic-map-based encryption in edge computing. The researchers employed several different analyses, such as secret key sensitivity analysis, encryption speed analysis, entropy analysis, and statistical randomness analysis. These analyses were conducted to showcase the security and effectiveness of the algorithm they put forward. The results of the differential attack revealed that the techniques proposed in the algorithm promote diffusion, which signifies resistance against statistical attacks. Such an encryption approach can remarkably address the assurance of data privacy in an IoT environment.

Similarly, Jamshidi et al. [29] also surveyed the applications of DL methods in response to COVID-19, but more from the perspective of diagnosis and treatment. More specifically, the paper covered and illustrated the applicability of DL methods such as Generative Adversarial Networks (GANs), Extreme Learning Machines (ELM), and Long/Short Term Memory (LSTM). The paper summarises some of the medical platforms that use AI and DL methods for deep analysis of medical images and rapid identification of diseases. A particular example is by using a layered approach, starting from the input layer database, selection, image-based techniques, optimisation, and finally, the output that produces the outcome of the final diagnosis. The paper concluded that AI and DL-based platforms can also accelerate the process of diagnosis and treatment of the COVID-19 disease, and discussed the different parameters surrounding DL methods, including properly choosing the inputs and targets of the neural network.

In a separate work, Rahman et al. [30] also investigated a B5G-based tactile edge learning method for handling COVID-19. This framework leveraged edge computing with 5G Radio Access Networks (RAN) by deploying a hierarchical edge computing architecture (ensuring scalability, low latency, and privacy) of data training and analysis. The proposed framework uses a distributed DL approach where each edge node runs its own local DL framework in conjunction with a three-phase reconciliation with the global DL framework in the cloud. Semantics are added to DL models to enable human experts to gain insights and semantic visualisation to facilitate key decision-making activities. The authors implemented a subset of COVID-19 scenarios using the distributed DL framework and demonstrated the effectiveness of the system.

In [31], Nguyen et al. provided a comprehensive review of AI methods in fighting the COVID-19 outbreak and outlined several critical research challenges in furthering AI research to better respond to the COVID-19 pandemic. This paper outlined the key roles, technical advantages, and challenges of AI, ranging from medical image processing, data analytics, text mining and natural language processing and the IoT, to computational biology and medicine. Furthermore, the authors also provided a summary of reliable COVID-19 data sources for research purposes, which can be valuable for the research community. The authors identified future research directions and highlighted specific groups of problems of AI in COVID-19, with the intention of providing the wider community with a comprehensive overview and motivating researchers in utilising AI to combat a global pandemic such as COVID-19.

An important application of IoT in the fight against COVID-19 is the use of drones or Unmanned Aerial Vehicles (UAVs). In [32], Kumar et al. investigated various drone-based systems and architectures, and discussed how these systems can effectively address the different challenging scenarios brought about by the COVID-19 pandemic. The proposed architecture involves the use of wearables in Body Area Networks (BANs) and implemented drone-based systems to collect a large amount of data across a large area. These data collection and analysis can be applied to use cases such as large-area sanitisation, thermal image collection, and patient identification within a short period of time. The work also outlined open questions and research challenges for drone-based technologies.

Chamola et al. [33] also provided a comprehensive survey on the use of UAVs, IoT, blockchain, AI, and 5G to mitigate the impact the COVID-19. The authors discussed in detail the Internet of Medical Things (IoMT), which is the collection of medical devices and software applications that are connected to the larger healthcare systems. The IoMT

is increasingly prevalent with an increasing number of mobile devices being equipped with Near Field Communication (NFC) and Bluetooth capabilities. IoMT can enable the monitoring of patients from a remote location, the tracking of medical orders, and the use of wearables to transmit healthcare data. Similar to IoT, IoMT enables the effective collection, analysis, and transmission of large amounts of healthcare data. The authors also discussed the roles of drones, robots, autonomous vehicles, wearables, telemedicine, mobile apps, blockchain for distributed databases, and 5G for thermal imaging. The different types of drones, such as multi-purpose drones, thermal imaging drones, announcement drones, disinfectant drones, surveillance drones, and medical drones are also discussed.

Gupta et al. [34] discussed in detail how blockchain can be used to facilitate UAVs in response to the COVID-19 pandemic. The authors proposed multi-swarm UAVs with 5G systems that aim to reduce human intervention (and reduce the risks of infections) while performing various critical operations. The UAV system utilised the advantages of 5G to send a large amount of data to ground stations in real-time, which enabled 5G's high bandwidth, high availability, and low latency properties. The blockchain component of the framework addresses the security and privacy issues when sending sensitive data across the network, which can be coupled with the upcoming innovations in 6G for using Terahertz (THz) frequency bands, the virtualisation of link and physical-layer protocols, and the softwarisation of communication infrastructure. Their studies showed that the proposed system achieved improved performance in terms of processing delay, packet losses, and throughput when compared with existing 4G/5G systems.

In [35], Siriwardhana et al. provide an overview of how 5G and IoT can be fully utilised in the battle against the COVID-19 pandemic. More specifically, the authors identified seven use cases, along with their technical requirements, that can benefit from the latest advances in 5G and IoT, namely telehealth, contact tracing and self-isolation, online education, retail and supply chain, smart manufacturing and factory automation, e-government and media, e-tourism, and entertainment. The paper also discussed how the advancements and developments in these areas can benefit society in a post-pandemic world.

The state-of-the-art IoT solutions discussed in the related work and presented in Table 1 primarily concentrate on implementing device-to-cloud IoT architecture. This preference stems from the advantages offered by cloud-based IoT architecture, such as scalability, advanced analytics, data storage and management capabilities, integration with external services, and remote monitoring and management features. However, it is important to acknowledge that relying solely on a cloud platform can have a detrimental effect on meeting the time constraints of time-sensitive IoT (TS-IoT) applications.

The process of transferring data from the IoT device layer to the cloud inherently introduces latency, which can vary depending on the type of data and the capacity of the communication network. This latency, as well as the possibility of data loss during data transfer from IoT devices to the cloud, can significantly impact IoT applications where meeting time constraints is of utmost importance. In such scenarios, a device-centric architecture such as the Sazgar framework can address these issues.

The Sazgar framework utilises approximation techniques to perform data processing at the first layer of the IoT architecture, which is the IoT device layer. By adopting this approach, the negative effects of latency and data loss on TS-IoT applications can be mitigated or eliminated.

# 3. Scope and Requirements of COVID-19 Symptoms and Compliance Monitoring Applications

The effective monitoring of citizens' compliance with COVID-19 lockdown restrictions is important for governments and municipalities across the world as this can provide important information that they need to ease or tighten lockdown restrictions, as well as formulate related policies and action plans for mitigating COVID-19 transmission. However, to be effective, COVID-19 citizen compliance monitoring must be both accurate and timely. More specifically, IoT applications for monitoring citizen compliance to COVID-19 restrictions must accurately and in a timely manner sense and compute the following high-value information for citizen activities in public places, such as buses, train carriages, bus stops, playgrounds, and food stores: (1) the real-time occupancy of public places; (2) social distancing in public places; (3) wearing masks; (4) COVID-19 symptoms, i.e., high body temperature and coughing; and (5) related citizen contact networks (e.g., the network of all passengers who happen to ride the same bus or the citizens that shopped in the same supermarket) without violating citizen privacy. In this paper, we consider an IoT application for monitoring COVID-19 citizen compliance. More specifically, we consider how to accurately and in a timely manner sense and compute the following:

- Social distancing: The World Health Organisation (WHO) and governments around the world recommend that during the pandemic citizens should maintain at least 1.5 m social distancing in public places [36]. Physical greetings such as handshaking, hugs, and kisses should be avoided. The COVID-19 application can use image processing and the relevant classifier on RGB camera data to estimate the distance of people in a public place.
- Wearing mask: The WHO, and some governments around the world, require health workers and citizens to wear a mask in the workplace or in public places [37,38].
- Curfew: Some governments around the world have imposed city-wide curfews to control the pandemic. For example, Melbourne in Australia instituted a 9 p.m. to 5 a.m. curfew for several weeks.
- Visiting hotspots: Some governments have declared hotspots in areas with a disproportionately greater than usual number of COVID-19 infections. This requires tracing citizens entering and leaving hotspots.
- People counting: Government restrictions often include limits on the number of people in specific public spaces, e.g., supermarkets.
- Symptoms: Monitoring COVID-19 symptoms (e.g., fever and coughing) is common in public places such as stations, schools, and workplaces.
- Contact networks: Many governments recommend that citizens use contact tracing applications on their mobile phones to assist in COVID-19 mitigation.

To provide COVID-19 compliance monitoring, the above citizen activities, symptoms, and contact networks must be detected within specific time-bounds and with specific accuracy to ensure the validity of such information.

Table 2, provides sample detection time-bound and corresponding accuracy required for detecting each of the above citizen activities and symptoms [37,39–45]. Although the detection accuracy for some of the activities such as people counting is relatively low, this data is only used as a basis of framework presentation. If detection of any activity or symptom occurs after its time-bound or with less accuracy than specified in Table 2, the COVID-19 compliance monitoring is ineffective.

Table 2. Detection time-bounds and corresponding accuracy for citizen activities and symptoms.

Citizen Activities and Symptoms in Public Places	<b>Detection Time-Bound</b>	<b>Detection Accuracy</b>
Social distancing	Less than 5 s	More than 65%
Wearing masks	Less than 30 s	More than 80%
Sanitising hands	Less than 5 s	More than 50%
Curfew	Less than 1 h	More than 95%
Visiting hotspots	Less than 30 min	More than 85%
People counting	Less than 3 s	More than 40%
Symptoms monitoring	Less than 30 s	More than 90%
Contact tracing	Less than 1 h	More than 95%

As described in Table 2, the main challenge in this IoT application is to achieve the appropriate balance between timelines and accuracy, in sensing and computing the above citizen activities and symptoms.

# 4. Device-Centric Approach in Processing TS-IoT Applications Using Approximation

Sazgar IoT proposed in this paper can meet the requirements of the COVID-19 compliance and symptoms monitoring application as well as other similar TS-IoT applications. To achieve this, Sazgar IoT incorporates a combination of an IoT-device-centric architecture with approximation-based IoT data analysis techniques. Data approximation is defined in Definition 2.

**Definition 1.** *Contextualisation is defined as "the process of filtering, aggregating, and inferring IoT data by using relevant information to the applications using the data"* [1].

**Definition 2.** Data Approximation is using contextualisation to reduce data in terms of volume, variety, and velocity to deal with computational resource constraints while satisfying the application requirements (e.g., time-bound, accuracy, etc.).

To further explain these contributions, we first describe how Sazgar IoT processes TS-IoT applications in Section 4.1. Section 4.2 presents the device-centric architecture of Sazgar IoT, provides an example of a TS-IoT application execution in an IoT device, and discusses its benefits and limitations. Finally, in Section 4.3, we present the principles of approximation-based data analysis and discuss how such techniques address the limitations of device-centric architecture we introduced in Section 4.1.

#### 4.1. TS-IoT Application Processing in the Device-Centric Framework

Figure 2 shows how Sazgar IoT processes TS-IoT applications. More specifically, as we discussed in Section 3, TS-IoT applications are submitted to Sazgar IoT as a set of (possibly interdependent) tasks, which include the task source code that was provided by the TS-IoT application developers, and a TS-IoT application accuracy and time-bound specification file that defines the application's accuracy and time-bound requirements for each of the tasks that comprise the application. Figure 2 shows the submission steps of the accuracy and time-bound tasks' specification files of the TS-IoT application by arrows labelled with '1' and '2', respectively. Table 2 in Section 3 provides an example of the accuracy and time-bound requirements of the COVID-19 symptoms and compliance application. The accuracy and time-bound specification format for the TS-IoT application are discussed further in Section 5.1.



Figure 2. Device-centric approach in processing TS-IoT applications in Sazgar IoT.

IoT device providers that make IoT devices available to Sazgar IoT independently submit their approximation parameters to the framework. Figure 2 depicts the submission of the approximation parameters by IoT device providers with the arrows labelled '3'. This can take place at any time. However, the Sazgar IoT framework considers only the IoT

devices whose approximation parameters have been submitted by the point in time steps '1' and '2' as completed. The details of the IoT device approximation parameter format are presented in Section 5.2.

When a TS-IoT application task cannot meet its TS-IoT application requirements, the Sazgar IoT framework uses the approximation techniques we discuss in Section 4.3 to tailor the IoT device parameters and/or the task's sources code in a way that the task can meet its time-bound and accuracy requirements when it is executed in its target IoT device. The submission of the tailored approximation parameters/task code is depicted in Figure 2 by the arrow labelled '4'. Approximation techniques are explained later in this section. The IoT data that results from the task execution in the IoT devices (which we refer to them as task reports) to the Sazgar IoT framework is shown with the label '5' in Figure 2.

#### 4.2. Device-Centric Architecture

As illustrated in Figure 2, Sazgar IoT uses the available IoT devices that generate the IoT data required by each TS-IoT application to process its tasks.

This device-centric architecture eliminates all IoT data harvesting-related communication delays, and it may only require concise task reports to be sent to a designated IoT device or the cloud. To explain this further, consider the 'People counting', 'Wearing masks', and 'Social distancing' tasks of the COVID-19 symptoms and compliance application in Table 2. Sazgar IoT meets the time-bound and detection accuracy requirements of these tasks by harvesting and analysing IoT data directly from the IoT sensors connected to the IoT device. This data is utilised to compute the results for each task.

The device-centric IoT architecture is cheaper and faster to implement as it does not require edge computers to harvest IoT data. More importantly, it is not subject to related network delays from remote harvesting of IoT data. Therefore, the device-centric architecture of Sazgar IoT allows meeting the time-bound requirements of the COVID-19 compliance application in Table 2 by:

- avoiding the transmission of large, high-velocity IoT data streams to the cloud and/or edge computers that are required by existing device-to-cloud and device-to-edge-tocloud approaches;
- eliminating related network delays (e.g., some cloud-based IoT platforms suffer from 300+ ms network delays that make meeting the requirements in Table 2 impossible);
- not requiring a large-scale (e.g., city or country-wide) deployment of edge computers that is very costly and will take significant time to implement.

#### 4.3. Principles of Approximation-Based Data Analysis

The main benefit of the device-centric architecture employed by Sazgar IoT is that it eliminates all communication delays in the IoT data harvesting performed by TS-IoT applications. However, IoT devices typically have limited onboard computing resources (i.e., microprocessors with relatively small memory and storage) that may not be enough to compute their stateful or even stateless data analysis in such a way that TS-IoT tasks meet their time-bounds and accuracy requirements. To address this challenge, Sazgar IoT incorporates innovative approximation techniques for TS-IoT tasks.

The approximation techniques proposed in this paper take as input: (1) the task source code and time-bound specification files of each TS-IoT application, and (2) the descriptions of available computing resources in the IoT device that is targeted for processing each task. The approximation techniques then use this information to tailor the IoT data harvesting rate or/and accuracy so that the TS-IoT application tasks can be computed within its required time-bound and accuracy. More specifically, when a TS-IoT task *t* needs to be processed at the target IoT device *d*, Sazgar IoT uses either or both of the following approximation techniques:

• IoT device reconfiguration: This technique reconfigures the IoT device *d* so that the computing resources needed to harvest the IoT data harvested by *t* are less than the resources that are currently available in *d*;

• IoT data harvesting method substitution: This technique generates an IoT data harvesting method for *t* and it substitutes the application-provided IoT data harvesting method in *t* so *t* uses less computing resources than those available in *d*.

To provide examples of both these approximations techniques, please consider the task that detects social distancing in Table 2. The IoT device reconfiguration technique adds a configuration method to this task that reduces the frame rate and/or resolution of the IoT data from the onboard camera(s) so that the IoT data harvested for the social distancing task can be processed with the available computing resources there and at the same time distancing violations involving citizens in the camera view is detected with 65% accuracy. The IoT data harvesting method substitution technique achieves a balance between the computing resources in the target IoT device by substituting the IoT data harvesting method in the social distancing task with a method that only measures the distance between a specific number of samples of citizens per minute (e.g., as they are entering and leaving the bus).

To automatically determine the degree of approximation required for each specific task and IoT device pair, we use contextual information (e.g., the available computing resources in each IoT device and TS-IoT application requirements, such as those in Table 2. Specific parameters used to perform the approximation include:

- IoT device computing resource parameters, including the IoT device's CPU and RAM usage;
- TS-IoT task IoT data harvesting parameters, including resolution of the video, colour profile (e.g., RGB, grey-scale, black-and-white), frame rate, and classification algorithm (e.g., depth or RGB).

Sazgar IoT uses these parameters to estimate (1) the accuracy of the task's results (i.e., the passenger counting accuracy in the bus), and (2) the CPU and RAM usage in the target IoT device.

Each TS-IoT application (e.g., IoT COVID-19 compliance) comprises several tasks (e.g., people counting, social distancing, etc.). Each of these tasks has a corresponding source code that is an executable description of the tasks and is provided by the TS-IoT application. These tasks can be divided into sub-tasks (e.g., video pre-processing) as illustrated in Figure 3.



Figure 3. Example of tasks and sub-tasks in TS-IoT applications.

Each of these sub-tasks may need to perform one of the following: consume IoT data from heterogeneous IoT devices, perform data processing ranging from basic stream processing to resource-intensive ML and statistics, manage the data queues required for stateful data analysis, and produce information that is used by other sub-tasks or reported to the users.

For example, in the people counting task we have video pre-processing, video processing, and people counting reporting tasks. Video processing involves running the Haar Cascade classifier [46,47] on the video to detect human heads that need to be counted. Video pre-processing is for preparing the data for the classifier. A few examples of pre-processing sub-tasks are illustrated in Figure 4. This video pre-processing is required for the classifier to reduce the complexity of the processing of the video. Reporting involves the mathematical summation of the heads and reporting the final result (i.e., the number of people).





The sub-tasks are not necessarily the same for all the tasks. For example, the temperature monitoring task will involve next face detection, infrared temperature processing, and providing citizen temperature reports in contrast with video pre-possessing, video classifier, and people counting report in people counting task as shown in Figure 3. All of these sub-tasks rely only on IoT devices to complete all required IoT data analysis tasks. This IoT-device-based analysis eliminates network delays from transferring high volume and velocity IoT data streams.

The TS-IoT application should provide the source code to Sazgar IoT to enable capturing accuracy, resource consumption, and time required to perform each of these sub-tasks. Sazgar IoT is responsible for performing the approximation and contextualisation to make sure the IoT application will meet the time-sensitivity and accuracy requirements.

The Sazgar IoT framework is described in detail in Section 5.

#### 5. Sazgar IoT Framework

In this section, we will discuss the Sazgar IoT framework. As illustrated in Figure 5, the Sazgar IoT framework consists of four main engines as follows:

- Approximation Context;
- Approximation;
- Task compiler and Deployment;
- Monitoring.

In the following section, we discuss the purpose and functionality of each of these engines. More specifically, the Approximation Context engine is presented in Section 5.1. The Approximation and Task Compiler and Deployment engine are discussed in Sections 5.2 and 5.3, respectively, while the Monitoring engine is introduced in Section 5.4.

The IoT application tasks can have input from one or more IoT devices. In this paper, we assume the data is coming from one IoT device. However, it is possible to decompose the tasks into sub-tasks and distribute them among different IoT devices (e.g., the IoT devices that are contributing the majority of the data for the task), which is outside of the scope of this paper.



Figure 5. Sazgar IoT framework.

#### 5.1. Approximation Context Engine

The Approximation Context engine is responsible for collecting the application-specific requirements of the TS-IoT application (we refer to this as application context) in the form of time-bound and accuracy pairs for each of the application's data analysis tasks and sending these specifications to the Approximation engine (discussed in Section 5.2) that (if needed) uses the specified time-bound and accuracy of each task to approximate it.

A TS-IoT application consists of several data analysis tasks. For example, the COVID-19 IoT application may consist of data analysis tasks such as social distance monitoring tasks, face masks wearing monitoring tasks, passenger counting tasks, etc. Such data analysis tasks have an accuracy and a time-bound requirement that need to be met during the execution.

For example, the requirement of the COVID-19 citizen compliance monitoring application in Table 2 is captured in an application specification file that is illustrated in Figure 6. It includes a time-bound and accuracy pair for each of the tasks in this application. The Approximation Context engine expects application specification files to be in XML format as illustrated in Figure 6 for the COVID-19 citizen compliance monitoring application tasks. Please note that the specified accuracy in Figure 6 is in percentage and the time-bounds are in seconds. Furthermore, each data analysis task comprises a set of actors that work together to complete the data analysis. For example, a passenger counting data analysis task may involve actors such as video pre-processing, video processing, and people count reporting.

In addition to the above specifications of TS-IoT applications, the Approximation Context engine periodically collects information about the available computing resources (i.e., the CPU usage, memory usage and execution times, etc.) that is provided by the Monitoring engine (discussed in Section 5.4) and forwards this information (we refer to this as computing resource context) to the Approximation engine, which uses this information to keep task approximations computable within the available computing resources at each relevant IoT device. More specifically, the computing resource context includes: (1) the available computing resources at the relevant IoT device (e.g., available processing, memory, and storage), and (2) the resources consumed by each running data analysis task (e.g., CPU usage, memory usage and processing time) for the same or other TS-IoT applications.

1	<covid_19></covid_19>
2	<social_distancing></social_distancing>
3	<time-bound>5</time-bound>
4	<accuracy>65</accuracy>
5	
6	<wearing_masks></wearing_masks>
7	<time-bound>30</time-bound>
8	<accuracy>80</accuracy>
9	
10	<sanitising_hands></sanitising_hands>
11	<time-bound>5</time-bound>
12	<accuracy>50</accuracy>
13	
14	<curfew></curfew>
15	<time-bound>3600</time-bound>
16	<accuracy>95</accuracy>
17	
18	<visiting_hotspots></visiting_hotspots>
19	<time-bound>1800</time-bound>
20	<accuracy>85</accuracy>
21	
22	<people_counting></people_counting>
23	<time-bound>3</time-bound>
24	<accuracy>40</accuracy>
25	
26	<symptoms_monitoring></symptoms_monitoring>
27	<time-bound>30</time-bound>
28	<accuracy>90</accuracy>
29	
30	<contact_tracing></contact_tracing>
31	<time-bound>3600</time-bound>
32	<accuracy>95</accuracy>
33	
34	

**Figure 6.** Accuracy and time-bound specifications for COVID-19 citizen compliance monitoring application.

This solution allows the Approximation Context engine to deal with IoT data rate and device volatility. In particular, in the presence of more volatility, the engine increases the rate of collecting computing resource context and forwards it to the Approximation engine. In the rest of this paper, we refer to both application context and computing resource context as context as contextual information [1].

#### 5.2. Approximation Engine

As its name implies, the Approximation engine uses the contextual information that is provided by the Approximation Context engine to tailor tasks of submitted TS-IoT applications in a way that each data analysis task can be fully executed with the computing resources available in a target IoT device while meeting its specified time-bounds and accuracy.

The Approximation engine receives two inputs including (1) Contextual information provided by the Approximation Context engine and (2) IoT device approximation parameters provided by the IoT device. IoT device approximation parameters are the parameters that can be modified to perform approximation. For example, Figure 7 shows that the IoT device has the capability to support three frame rates per second. IoT device approximation parameters are described with Semantic Sensor Network (SSN) ontology [48] and are described in more detail in Section 5.2.1. The Approximation engine accomplishes this by performing the following:

15 of 29

1	<owl:class rdf:about="#RGBCameraMeasurementFrequency"></owl:class>
2	<rdfs:label>RGBCamera measurement frequency</rdfs:label>
3	<rdfs:subclassof rdf:resource="http://purl.oclc.org/NET/ssnx/ssn#Frequency"></rdfs:subclassof>
4	<rdfs:comment>This class describes output frequency values of the RGB camera: Low (10Hz) ,</rdfs:comment>
	medium (20Hz), high (30Hz).
5	
6	<pre><owl:thing rdf:about="#RGBCameraMeasurementFrequencyValue"></owl:thing></pre>
7	<rdf:type rdf:resource="#FrequencyValue"></rdf:type>
8	<hasquantityvalue rdf:datatype="&amp;xsd;float">30</hasquantityvalue>
9	<rdfs:comment>Highest Resolution</rdfs:comment>
10	
11	<pre><owl:thing rdf:about="#RGBCameraMeasurementFrequencyValue"></owl:thing></pre>
12	<rdf:type rdf:resource="#FrequencyValue"></rdf:type>
13	<hasquantityvalue rdf:datatype="&amp;xsd;float">20</hasquantityvalue>
14	<rdfs:comment>Medium Resolution</rdfs:comment>
15	
16	<pre><owl:thing rdf:about="#RGBCameraMeasurementFrequencyValue"></owl:thing></pre>
17	<rdf:type rdf:resource="#FrequencyValue"></rdf:type>
18	<hasquantityvalue rdf:datatype="&amp;xsd;float">10</hasquantityvalue>
19	<rdfs:comment>Lowest Resolution</rdfs:comment>
20	

Figure 7. IoT device approximation parameters.

- 1. Constructs the decision graph that will be described in Section 5.2.1. The decision graph construction aim is to capture IoT device approximation parameters provided by IoT devices and will use those parameters to provide resource consumption for all the possible combinations of IoT device approximation parameters as well as expected accuracy and time-bounds for computing the tasks with those corresponding parameters.
- 2. Contextualise the decision graph that will be described in Section 5.2.2. It uses the constructed decision graph as an input to construct the contextualised decision graph. This contextualised decision graph is a contextual filter of the decision graph [49]. The output of this technique is a subset of the constructed decision graph where it only includes paths that can be computed with the available computing resources (provided by Computing Resource Context) and also satisfy the application specifications in terms of accuracy and time-bounds. For example, for the COVID-19 citizen compliance monitoring application we will have only paths that can satisfy the accuracy and time-bounds mentioned in Table 2 while they can be computed with the currently available computing resources.
- 3. Determining the most suitable Task Processing Path that will be described in Section 5.2.3. This technique will use a contextualised decision graph as an input and will select the paths that can achieve the highest accuracy to be provided to Task Compiler and Deployment engine.
- 4. Uses computing resources context updates from the approximation context engine to dynamically contextualise the decision graph during the run-time.

In the following sections, we present the techniques we developed for constructing contextualised decision graphs and determining the most suitable Task Processing Paths.

# 5.2.1. Decision Graph Construction Technique

This technique constructs a directed acyclic graph, which we refer to as a decision graph, before the start of the application execution. To facilitate this, each available IoT device provides IoT device Approximation Parameters to the Approximation engine. These specifications describe the various approximation parameters such as IoT data type (e.g., RGB, Depth), number of frames (e.g., 10 frames), and so forth involved in the TS-IoT application. The "IoT device approximation parameter" follows SSN ontology concepts [48]. IoT device approximation parameters in this paper are described in the 'MeasuringCapability' module of the SSN as illustrated in Figure 8.



**Figure 8.** IoT device approximation parameters conceptual description utilising the Semantic Sensor Network Ontology.

Figure 7 represents an example of IoT device approximation parameters describing that the device has an RGB camera that supports three different frame rates including Low (10 fps), Medium (20 fps), and High (30 fps).

The decision graph construction technique includes a cloud-based simulator that can be configured according to the system configurations of the underlying IoT device by the users of Sazgar IoT. As noted before, each data analysis task of the TS-IoT application comprises a set of actors. Each actor of the data analysis task has a corresponding sample actor (i.e., an actor that can be simulated in the simulator) and a source code actor (i.e., an executable unit of the framework that is a compiled source code). By using this simulator and IoT device approximation parameters, before the actual deployment and execution of the IoT application, first, we simulate the sample actors of the data analysis tasks under each of the IoT device approximation parameters specified in the file and observed. Each of these simulations under different IoT device approximation parameters will lead to an observation of accuracy, resource consumption (observed from the simulator), and total application execution time. We refer to these observations as Task Processing Paths. After the simulations, by using these Task Processing Paths, the technique constructs a contextualised decision graph.

# 5.2.2. Contextualised Decision Graph Technique

The contextualised decision graph would define the approximation of the IoT processing for each application data analysis task (i.e., video frame rates, video resolution, etc.) which leads to different resource usages, accuracy levels, and total application execution times. As we discussed earlier, each data analysis task comprises a set of actors and each actor has its own IoT device approximation parameters (e.g., sample actor variation for frame size IoT device approximation parameter may involve 10 frames actor, 20 frames actor, and 30 frames actor). This is only performed at the beginning of the actual application deployment. The constructed decision graph will then be used to identify the most suitable Task Processing Path that meets the time-bound requirements of the application based on the contextual information.

The decision tree construction technique has the capability to update the Task Processing Paths based on the resource consumption information of the executed actors in cases where the observations (i.e., resource consumption, total application execution time) are not matching the simulation output. To realise this, the decision tree construction technique utilises the computing resource context information (mainly the resource consumption information of the actors), which is periodically sent by the approximation context engine. In addition, this technique can add to the Task Processing Paths by executing sample actor variation in IoT device, edge, or cloud to estimate accuracy, computing resource context, and time-bound for the actors not provided by the simulator.

#### 5.2.3. Task Processing Path Selector Technique

This technique is responsible for selecting the most suitable actor processing for each of the data analysis tasks that can meet the time-bound requirements of the TS-IoT application. A Task Processing Path is a single end-to-end path of a data analysis task in the contextualised decision graph. For example, in our use case, the video preprocessing actor (e.g., grey-scale)  $\rightarrow$  video processing actor  $\rightarrow$  people count report actor is a sample Task Processing Path of the passenger counting data analysis task. Before the actual deployment of the TS-IoT application, the Task Processing Path selector receives the contextualised decision graph that was constructed with the help of the simulator. Then, the Task Processing Path selector technique uses the application requirements context received from the approximation context engine to select the most suitable Task Processing Path that leads to an observation of the highest accuracy within the time-bound requirements specified in the application requirements context for each individual task. The selected Task Processing Paths will be forwarded to the Task Compiler and Deployment engine which will then manage the deployment of the IoT application into the IoT device. This will be discussed in Section 5.3.

During the run-time of the TS-IoT application, the approximation context engine periodically collects computing resource context from the monitoring engine, and this information is sent to the Approximation engine. Whenever the observed total application execution time is beyond the time-bound requirements of the TS-IoT application, the Task Processing Path selector disables the existing Task Processing Path and dynamically finds a suitable Task Processing Path that has the next highest accuracy. In order to achieve this, the engine uses a "Dynamic decision graph-based algorithm", which is discussed in Section 5.2.4. Then, this Task Processing Path is forwarded to the Task Compiler and Deployment engine for a redeployment.

#### 5.2.4. Dynamic Decision Graph-Based Algorithm

In our proposed solution, we observe the resource consumption and execution time of the actors under different IoT device approximation parameters. Each execution of an atomic process will lead to an observation of execution time and consumed resources. We then use these observations to update the constructed decision graph, which defines the approximation of the IoT processing for each application task (i.e., video frame rates, video resolution, type of classification algorithm used, etc.), which leads to different resource usages, accuracy levels, and execution times [Figure 9].

The constructed decision graph can be used to identify the most suitable Task Processing Path (i.e., suitable frame rate, video resolution) that would help meet the application's time-sensitive requirements while (at least) achieving an accuracy level defined by the application.

In our proposed Task Processing Path selection algorithm, we dynamically identify the most suitable Task Processing Path, whenever the contextual information changes. The proposed algorithm continuously checks whether the contextual information is updated or not. Whenever the contextual information is updated, it collects that contextual information and passes it to the "Approximate-IoT-Data" function. The approximate-IoT-Data function is executed in the Approximate engine. The function first takes the approximation decision graph, which was constructed initially with the simulation. Subsequently, the algorithm contextualise the decision graph using the computing resource context. Here, the Task Processing Paths in the decision graph are updated based on the resource consumption information of the source code actors in cases where the observations (i.e., resource consumption, total application execution time) do not match the simulation output. After that, the algorithm finds the most suitable Task Processing Path that yields the highest accuracy within the time-bound requirements of the application using the application requirements context. This "actor-process-path" is then forwarded to the Actor construction engine via the Create-Actors function. The Create-Actors function constructs the source code actors and deploys them into the IoT devices. Our dynamic Task Processing Path selection algorithm is shown in Algorithm 1.



Figure 9. Approximation Decision Tree Graph.

**Algorithm 1:** Pseudo-code for dynamically deciding the Task Processing Path, creating corresponding actors, and deploying them in IoT devices

Initialise Application Requirements and Resource Computing Context; Initialise Actor Decision Graph;

# while True do

- if (Contextual-information-is-updated) then
  - Application-Context # Read Application Requirements from IoT application;
  - Computing-Resource-Context # Read Computing Resource Context from monitored data;
  - Approximate-IoT-Data(Application-Context,
  - Computing-Resource-Context);

#### end

#### end

**Function** Approximate-IoT-Data(Application-Context, Computing-Resource-Context) is Decision-Graph = ContextualiseDecisionGraph(Actor Decision Graph,

- Computing-Resource-Context) # Contextualise the decision tree graph based on the Computing Resource Context;
- Actor-Process-Path = Select Path(Decision-Graph, Application-Context) # Find the Task Processing Path that yields in the highest accuracy within the given time-bound requirements;

Create-Actors (Actor-Process-Path)# Create Actors for the corresponding actors in the Task Processing Path;

# end

**Function** *Create-actors* (*Actor-Process-Path*) **is** 

Actor-List = Construct(Actor-Process-Path)# Deploy the Actors and start the execution; Deploy-Actors(Actor-List);

```
end
```

# 5.3. Task Compiler and Deployment Engine

This engine is mainly responsible for constructing the source code actors(i.e., executable units of the framework that are compiled source codes) that can run in the IoT devices and deploying them into the IoT device. To achieve this, the engine accepts the Task Processing Path selected by the approximation engine and constructs the corresponding source code actors for each actor in the Task Processing Path. Finally, the engine deploys the corresponding source code actor to the IoT device and starts its execution. Whenever the engine receives an updated Task Processing Path, it stops the execution of the running source code actor in the IoT devices and deletes them. After that, it constructs new source code actors according to the updated Task Processing Path and deploys them into the IoT device. The IoT device is responsible for executing each source code actor and collecting the data. In the scope of this paper, there is no distribution of the tasks between the IoT devices.

#### 5.4. Monitoring Engine

This engine is responsible for continuously monitoring the performance metrics of the underlying IoT device and source code actors. The monitoring engine periodically collects performance metrics such as available CPU, memory, and storage from the underlying IoT device. Moreover, it collects the resource consumption information such as CPU usage, memory usage, and processing time of each source code actor. Subsequently, the monitored data are fed into the approximation context engine.

For example, if the application requires an accuracy level of 80% within 10 s, then the approximation engine identifies the Task Processing Path which leads to an 80% level of accuracy within 10 s. The Task Processing Path may include an actor which has parameters such as RGB video, 70% video size, and 20 frame rates. Then, this will trigger an invocation to the actor construction engine, which will create the corresponding source code actor (with the respective parameters). These source code actors will be then deployed into the underlying IoT device and remove the previous source code actors. The monitoring engine will continuously monitor the performance metrics of resources and source code actors and feed this information into the approximation context engine. Therefore, the engine will have real-time resource usage information and is aware of the performance metrics of the source code actors as well (i.e., in terms of execution time).

When the contextual information is changed, the creation of a new set of actors will be triggered that are more suitable for achieving the time-sensitive requirements of the IoT application in the current context. Likewise, whenever the context is changed, the approximation context engine will forward this information to the Approximation engine, then the Approximation engine dynamically identifies the suitable Task Processing Paths. The Task Compiler and Deployment engine stops the execution of previous actors and creates the corresponding actors and deploys the newly created actors to the corresponding IoT device so that the time-sensitive requirement of the IoT application is met.

#### 6. Experimental Evaluation

In this section, we describe our experimental testbed setup and evaluation methodology. The experiments described in this paper were conducted on identical IoT devices with 1GB of RAM with a Linux Ubuntu v18.04 LTS operating system.

To conduct the experiments, we implemented a proof of concept of the framework's engines using Microsoft Orleans Actor framework [50]. Orleans actor framework is based on the mathematical concurrent computation model called the 'actor model'. In the actor model, an 'actor' is considered as the primitive unit of computation and it communicates with other actors through message passing. Orleans actors are developed to scale in an elastic way and then can be executed on any operating system which has a .NET core installed. Therefore, we decided to implement our framework's actors as Orleans actors and this facilitated the development of a highly scalable and efficient Task Compiler and Deployment engine on top of the Orleans software framework.

The computing resource including RAM and CPU usage, as well as the breakdown of processing time for each atomic process, have been monitored in real time. The monitoring samples were collected at one-second intervals.

The COVID-19 application algorithm used in the paper has two main components including video pre-processing and video processing (i.e., classification) algorithms.

Video pre-processing is a set of tasks to prepare the video to perform processing algorithms. These tasks can include, but are not limited to, colour profile modification, noise reductions, and so forth. A few examples of video pre-processing tasks are illustrated in Figure 4.

The processing algorithms are responsible for monitoring the enforcement compliance and symptoms monitoring. We utilised 'Haar Cascade' object detection algorithm [46,47]. A description of the computer vision algorithms for compliance enforcement and symptom monitoring is outside of the scope of this paper.

#### 6.1. Results and Analysis

In this section, the experimental results collected from the experiments will be presented and discussed. The experiments discussed in this paper were performed in a controlled environment using identical IoT devices. We will describe how Sazgar IoT can effectively support time-sensitive COVID-19 citizen compliance monitoring applications.

Figures 10 and 11 show the processing time for performing the processing (i.e., classification) algorithm (i.e., people detection and counting) on the RGB and depth video data, respectively. As can be seen from the figures, counting the number of people entering and exiting the buses utilising higher resolution video and higher frame rates still need significantly more time to perform the classification. Moreover, RGB video data require two times more time to process the classification on high-resolution videos as compared to depth video. The difference is not very significant for lower-resolution videos.



**Figure 10.** Comparison of the video processing (i.e., classification) time for different RGB video size (width in pixel) and frame rates (per second).



**Figure 11.** Comparison of the video processing (i.e., classification) time for different depth video size (width in pixel) and frame rates (per second).

Figures 12 and 13, show the resource consumption for the video pre-processing algorithm. Figures 14 and 15, show the CPU and RAM usage for the processing algorithm performed on RGB and depth video data. Table 3 compares the depth and RGB video pre-processing computing resource consumption. It can be observed that the resource consumption for the processing algorithm is higher than the video pre-processing sub-tasks. In addition, resource consumption follows a linear trend for higher video resolutions in such a way that higher-resolution videos always consume more resources to perform processing algorithms.



**Figure 12.** Comparison of the RGB video pre-processing computing resource consumption—RAM and CPU.



**Figure 13.** Comparison of the depth video pre-processing computing resource consumption—RAM and CPU.



**Figure 14.** Comparison of the RGB video processing (i.e., classification) computing resource consumption—RAM and CPU.

Video Width	Depth Frame 10 RAM %	CPU %	Frame 20 RAM %	CPU %	Frame 30 RAM %	CPU %	RGB Frame 10 RAM %	CPU %	Frame 20 RAM %	CPU %	Frame 30 RAM %	CPU %
64	2.06	1.81	3.59	3.17	5.62	5.53	1.58	1.72	3.74	3.69	4.76	4.65
128	2.19	1.86	2.78	2.65	5.20	5.15	1.63	1.70	3.94	3.84	5.19	5.07
192	2.46	2.03	3.79	3.67	4.16	4.64	1.67	1.75	3.41	3.58	4.68	4.70
256	2.65	1.98	3.96	4.17	5.67	6.12	2.33	2.41	3.43	3.69	4.89	4.84
320	1.86	1.61	3.77	3.77	3.76	4.07	2.37	2.44	4.05	3.89	4.45	4.42
448	3.59	3.07	6.84	6.88	6.59	7.35	2.08	2.33	6.82	6.34	7.15	6.93
512	2.21	2.11	8.20	8.08	6.62	7.28	3.23	3.43	6.19	6.05	7.81	7.58
576	3.08	2.73	6.86	7.03	3.44	4.36	2.41	2.75	6.23	6.13	5.98	6.41

Table 3. Comparison of the depth and RGB video pre-processing computing resource consumption.





It is obvious from the figures discussed earlier that the resource consumption and processing time are dependent on the type of videos, video resolution, frame rate, and type of algorithms performed on the data as illustrated in Figure 16.

In this paper, we propose a framework that can understand and take into account these dependencies and use them to perform more effective and efficient IoT data analysis on IoT devices within the time-sensitive requirements while respecting the available computing resources and preferences/requirements from the applications.

Figure 17, presents the processing algorithm's accuracy with respect to frame rate and video resolution (i.e., video width in pixels). The accuracy is represented with a percentage where a higher value means higher accuracy.

The blue surface illustrated in Figure 17 represents an assumed required/preferred accuracy by application. This surface is dynamic and any values above the surface will have sufficient accuracy to be selected by the proposed framework (e.g., in this example, acceptable accuracy is more than 40%; however, in real-world applications, the acceptable accuracy should be more than 70%).



Figure 16. Computation time comparison of depth and RGB people counter.



Figure 17. Comparison of the accuracy for different video and frame size.

Figure 18, represents the RAM and CPU usages for running processing algorithms on the depth video data. Similarly, the blue surface indicates the available/preferred computing resource to perform the processing task (e.g., utilising not more than 10% of the computing resource). The proposed framework chooses the frame rate and video size where the RAM and CPU consumption are both under the blue surface to fulfil the requirements/preferences.



Figure 18. Comparison of the resource consumption for different video and frame size.

Finally, Figure 19 shows the duration of the process with respect to the video size and frame rates. The blue surface in this figure represents the time-sensitive requirements (e.g., 15 s). As it is illustrated in the figure, any combination of the frame rate and video size in which they can perform the process within the time boundaries can be selected by our proposed framework.



Figure 19. Comparison of the duration of the processes for different video and frame size.

# 6.2. Limitations

While the experimental evaluation demonstrates that the proposed framework fulfils the time-bound and accuracy requirements of the monitoring application, there are certain limitations in the existing approach that need to be addressed. One such limitation is that the experiment was conducted using a fixed number of IoT devices, and it does not account for the potential impact of fluctuations in the number of IoT devices on the accuracy of the approximation results. In addition, the impact of IoT device interoperability on TS-IoT applications must be addressed if other types of sensors are added to IoT devices. Another limitation pertains to data availability. Although data availability is not a limitation of the proposed framework, it is an issue that can impact the accuracy of such time-bound frameworks. Therefore, it is crucial to prioritise data availability in TS-IoT applications that utilise Sazgar IoT to ensure the accuracy of the monitoring process. Furthermore, although device-centric IoT frameworks such as Sazgar IoT can significantly improve the efficiency of TS-IoT applications, it must also be considered that they compromise cloud-specific IoT capabilities such as scalability, advanced analytics, data storage and management, integration with external services, and remote monitoring and management. Another notable drawback of the suggested framework is that the approximation technique may not be suitable for IoT applications that necessitate a 100 percent accuracy rate. This implies that there is a trade-off between the time sensitivity and accuracy sensitivity of the IoT application, which must be taken into account when choosing between deviceto-cloud or device-centric architectures. While device-centric frameworks such as Sazgar can significantly enhance processing speed, they may also result in a lower accuracy rate compared to cloud-based architectures.

#### 7. Conclusions and Future Work

As the IoT brought about unprecedented changes in modern society, the need for collecting and processing data efficiently has become more important than ever. There exist many approaches to data harvesting and processing but many rely on highly resourceful devices such as those in the cloud. While many applications benefit from outsourcing data processing, time-sensitive applications that are sensitive to delays can benefit from processing the collected data locally (trading off computational resources with time). In this paper, we proposed an IoT device-centric data analysis framework for time-sensitive monitoring of COVID-19 lockdown compliance and symptoms. This framework can take into account contextual information and perform approximation to enable the processing of the data on the IoT device itself while satisfying the application-specified outcomes and time-sensitive requirements. The contextual information (Computing Resource Context and Application Requirement Context) and approximation technique subsequently enabled IoT end devices to intelligently perform several tasks relevant to monitoring COVID-19 lockdown compliance and symptoms on IoT devices without needing access to cloud or edge computing resources. It must be noted that when using the proposed framework in real-world scenarios, the ideal activity detection accuracy should be more than 70%.

Future work involves investigating accuracy-based decision making in such a way that the framework will dynamically switch to a processing path that delivers the best accuracy in any given context. Another avenue to explore would be the use of a granular and context-aware disclosure technique as proposed in [51] to maintain privacy-preserving in our COVID-19 application. This technique can dynamically maintain data security utilising the contextual information described in this paper. Other research directions include deploying rate adaptation intelligence at video-enabled end devices [52], implementing latency-reducing techniques [53], incorporating network-based feedback (bandwidth/delay measurements and predictions) as contextual information, and experimenting with advanced data analytic techniques and emerging IoT-based communication technologies.

**Author Contributions:** Conceptualization, A.Y.; Methodology, A.Y., H.K., D.G. and J.K.; Software, A.Y. and H.K.; Writing—original draft, A.Y., H.K. and J.K.; Writing—review & editing, D.G. and H.B.; Visualization, A.Y. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

**Acknowledgments:** This research was supported by use of the Nectar Research Cloud, a collaborative Australian research platform supported by the Australian Research Data Commons (ARDC). The ARDC is funded by the National Collaborative Research Infrastructure Strategy (NCRIS).

Conflicts of Interest: The authors declare no conflict of interest.

#### References

- 1. Yavari, A. Internet of Things Data Contextualisation for Scalable Information Processing, Security, and Privacy. Ph.D. Thesis, RMIT University, Melbourne, Australia, 2019.
- Korala, H.; Georgakopoulos, D.; Jayaraman, P.P.; Yavari, A. A Survey of Techniques for Fulfilling the Time-Bound Requirements of Time-Sensitive IoT Applications. ACM Comput. Surv. (CSUR) 2022, 54, 1–36. [CrossRef]
- 3. Korala, H.; Georgakopoulos, D.; Jayaraman, P.P.; Yavari, A. Managing time-sensitive iot applications via dynamic application task distribution and adaptation. *Remote Sens.* **2021**, *13*, 4148. [CrossRef]
- 4. Kashevnik, A.; Lashkov, I.; Gurtov, A. Methodology and Mobile Application for Driver Behavior Analysis and Accident Prevention. *IEEE Trans. Intell. Transp. Syst.* **2020**, *21*, 2427–2436. [CrossRef]
- Korala, H.; Jayaraman, P.P.; Yavari, A.; Georgakopoulos, D. APOLLO: A Platform for Experimental Analysis of Time Sensitive Multimedia IoT Applications. In Proceedings of the 18th International Conference on Advances in Mobile Computing & Multimedia (MoMM'20), Chiang Mai, Thailand, 30 November–2 December 2020; Association for Computing Machinery: New York, NY, USA, 2021; pp. 104–113. [CrossRef]
- Yavari, A.; Bagha, H.; Korala, H.; Mirza, I.; Dia, H.; Scifleet, P.; Sargent, J.; Shafiei, M. ParcEMon: IoT Platform for Real-Time Parcel Level Last-Mile Delivery Greenhouse Gas Emissions Reporting and Management. Sensors 2022, 22, 7380. [CrossRef] [PubMed]
- Korala, H.; Georgakopoulos, D.; Jayaraman, P.P.; Yavari, A. A Time-Sensitive IoT Data Analysis Framework. In Proceedings of the Hawaii International Conference on System Sciences (HICSS 54), Kauai, HI, USA, 5 January 2021; p. 7185. [CrossRef]
- Taneja, M.; Davy, A. Resource aware placement of IoT application modules in Fog-Cloud Computing Paradigm. In Proceedings of the 2017 IFIP/IEEE Symposium on Integrated Network and Service Management (IM), Lisbon, Portugal, 8–12 May 2017; pp. 1222–1228. [CrossRef]
- 9. Yousefpour, A.; Patil, A.; Ishigaki, G.; Kim, I.; Wang, X.; Cankaya, H.C.; Zhang, Q.; Xie, W.; Jue, J.P. FogPlan: A Lightweight QoS-aware Dynamic Fog Service Provisioning. *CoRR* 2018. [CrossRef]
- Brogi, A.; Forti, S. QoS-Aware Deployment of IoT Applications Through the Fog. *IEEE Internet Things J.* 2017, 4, 1185–1192. [CrossRef]
- Chandan, G.; Jain, A.; Jain, H.; Mohana. Real time object detection and tracking using Deep Learning and OpenCV. In Proceedings of the 2018 International Conference on Inventive Research in Computing Applications (ICIRCA), Coimbatore, India, 11–12 July 2018; pp. 1305–1308.
- 12. Pulli, K.; Baksheev, A.; Kornyakov, K.; Eruhimov, V. Real-time computer vision with OpenCV. *Commun. ACM* **2012**, 55, 61–69. [CrossRef]
- Grönman, J.; Sillberg, P.; Rantanen, P.; Saari, M. People Counting in a Public Event—Use Case: Free-to-Ride Bus. In Proceedings of the 2019 42nd International Convention on Information and Communication Technology, Electronics and Microelectronics (MIPRO), Opatija, Croatia, 20–24 May 2019; pp. 1055–1059.
- 14. Sun, S.; Akhtar, N.; Song, H.; Zhang, C.; Li, J.; Mian, A. Benchmark data and method for real-time people counting in cluttered scenes using depth sensors. *IEEE Trans. Intell. Transp. Syst.* **2019**, *20*, 3599–3612. [CrossRef]
- Nieto-Rodríguez, A.; Mucientes, M.; Brea, V.M. System for medical mask detection in the operating room through facial attributes. In Proceedings of the Iberian Conference on Pattern Recognition and Image Analysis, Santiago de Compostela, Spain, 17–19 June 2015; Springer: Berlin/Heidelberg, Germany, 2015; pp. 138–145.
- Skarlat, O.; Schulte, S.; Borkowski, M.; Leitner, P. Resource Provisioning for IoT Services in the Fog. In Proceedings of the 2016 IEEE 9th International Conference on Service-Oriented Computing and Applications (SOCA), Macau, China, 4–6 November 2016; pp. 32–39. [CrossRef]
- 17. Mirza, I.B.; Georgakopoulos, D.; Yavari, A. Cyber-Physical-Social Awareness Platform for Comprehensive Situation Awareness. *Sensors* **2023**, *23*, 822. [CrossRef]
- Michalák, P.; Watson, P. PATH2iot: A Holistic, Distributed Stream Processing System. In Proceedings of the 2017 IEEE International Conference on Cloud Computing Technology and Science (CloudCom), Hong Kong, China, 11–14 December 2017; pp. 25–32. [CrossRef]
- Bagha, H.; Yavari, A.; Georgakopoulos, D. Hybrid Sensing Platform for IoT-Based Precision Agriculture. *Future Internet* 2022, 14, 233. [CrossRef]

- Skarlat, O.; Nardelli, M.; Schulte, S.; Dustdar, S. Towards qos-aware fog service placement. In Proceedings of the 2017 IEEE 1st international conference on Fog and Edge Computing (ICFEC), Madrid, Spain, 14–15 May 2017; pp. 89–96.
- Yavari, A.; Jayaraman, P.P.; Georgakopoulos, D. Contextualised service delivery in the Internet of Things: Parking recommender for smart cities. In Proceedings of the 2016 IEEE 3Rd world forum on internet of things (WF-iot), Reston, VA, USA, 12–14 December 2016; pp. 454–459.
- 22. Li, L.; Li, S.; Zhao, S. QoS-aware scheduling of services-oriented internet of things. IEEE Trans. Ind. Inform. 2014, 10, 1497–1505.
- Nguyen, C.T.; Saputra, Y.M.; Van Huynh, N.; Nguyen, N.T.; Khoa, T.V.; Tuan, B.M.; Nguyen, D.N.; Hoang, D.T.; Vu, T.X.; Dutkiewicz, E.; et al. A comprehensive survey of enabling and emerging technologies for social distancing—Part I: Fundamentals and enabling technologies. *IEEE Access* 2020, *8*, 153479–153507. [CrossRef] [PubMed]
- 24. Nguyen, C.T.; Saputra, Y.M.; Van Huynh, N.; Nguyen, N.T.; Khoa, T.V.; Tuan, B.M.; Nguyen, D.N.; Hoang, D.T.; Vu, T.X.; Dutkiewicz, E.; et al. A comprehensive survey of enabling and emerging technologies for social distancing—Part II: Emerging technologies and open issues. *IEEE Access* **2020**, *8*, 154209–154236. [CrossRef] [PubMed]
- Hussain, A.A.; Bouachir, O.; Al-Turjman, F.; Aloqaily, M. AI techniques for COVID-19. IEEE Access 2020, 8, 128776–128795. [CrossRef] [PubMed]
- Pham, Q.V.; Nguyen, D.C.; Huynh-The, T.; Hwang, W.J.; Pathirana, P.N. Artificial intelligence (AI) and big data for coronavirus (COVID-19) pandemic: A survey on the state-of-the-arts. *arXiv* 2021, arXiv:2107.14040.
- Hossain, M.S.; Muhammad, G.; Guizani, N. Explainable AI and mass surveillance system-based healthcare framework to combat COVID-I9 like pandemics. *IEEE Netw.* 2020, 34, 126–132. [CrossRef]
- Raghunandan, K.; Dodmane, R.; Bhavya, K.; Rao, N.K.; Sahu, A.K. Chaotic-Map Based Encryption for 3D Point and 3D Mesh Fog Data in Edge Computing. *IEEE Access* 2022, 11, 3545–3554. [CrossRef]
- Jamshidi, M.; Lalbakhsh, A.; Talla, J.; Peroutka, Z.; Hadjilooei, F.; Lalbakhsh, P.; Jamshidi, M.; La Spada, L.; Mirmozafari, M.; Dehghani, M.; et al. Artificial intelligence and COVID-19: Deep learning approaches for diagnosis and treatment. *Ieee Access* 2020, *8*, 109581–109595. [CrossRef] [PubMed]
- Rahman, M.A.; Hossain, M.S.; Alrajeh, N.A.; Guizani, N. B5G and explainable deep learning assisted healthcare vertical at the edge: COVID-I9 perspective. *IEEE Netw.* 2020, 34, 98–105. [CrossRef]
- 31. Nguyen, T.T.; Nguyen, Q.V.H.; Nguyen, D.T.; Hsu, E.B.; Yang, S.; Eklund, P. Artificial intelligence in the battle against coronavirus (COVID-19): A survey and future research directions. *arXiv* 2020, arXiv:2008.07343.
- 32. Kumar, A.; Sharma, K.; Singh, H.; Naugriya, S.G.; Gill, S.S.; Buyya, R. A drone-based networked system and methods for combating coronavirus disease (COVID-19) pandemic. *Future Gener. Comput. Syst.* **2021**, *115*, 1–19. [CrossRef]
- 33. Chamola, V.; Hassija, V.; Gupta, V.; Guizani, M. A comprehensive review of the COVID-19 pandemic and the role of IoT, drones, AI, blockchain, and 5G in managing its impact. *IEEE Access* 2020, *8*, 90225–90265. [CrossRef]
- Gupta, R.; Kumari, A.; Tanwar, S.; Kumar, N. Blockchain-envisioned softwarized multi-swarming uavs to tackle covid-i9 situations. *IEEE Netw.* 2020, 35, 160–167. [CrossRef]
- Siriwardhana, Y.; De Alwis, C.; Gür, G.; Ylianttila, M.; Liyanage, M. The fight against the COVID-19 pandemic with 5G technologies. *IEEE Eng. Manag. Rev.* 2020, 48, 72–84. [CrossRef]
- 36. World Health Organization. *Coronavirus Disease 2019 (COVID-19): Situation Report,* 72; World Health Organization: Geneva, Switzerland, 2020.
- Feng, S.; Shen, C.; Xia, N.; Song, W.; Fan, M.; Cowling, B.J. Rational use of face masks in the COVID-19 pandemic. *Lancet Respir.* Med. 2020, 8, 434–436. [CrossRef] [PubMed]
- Greenhalgh, T.; Schmid, M.B.; Czypionka, T.; Bassler, D.; Gruer, L. Face masks for the public during the COVID-19 crisis. *Bmj* 2020, 369, m1435. [CrossRef]
- Feng, Y.; Marchal, T.; Sperry, T.; Yi, H. Influence of wind and relative humidity on the social distancing effectiveness to prevent COVID-19 airborne transmission: A numerical study. *J. Aerosol Sci.* 2020, 147, 105585. [CrossRef]
- Zhao, T.; Cheng, C.; Liu, H.; Sun, C. Is one-or two-meters social distancing enough for COVID-19? Evidence for reassessing. *Public Health* 2020, 185, 87. [CrossRef]
- 41. Matrajt, L.; Leung, T. Evaluating the effectiveness of social distancing interventions to delay or flatten the epidemic curve of coronavirus disease. *Emerg. Infect. Dis.* **2020**, *26*, 1740. [CrossRef] [PubMed]
- 42. Punn, N.S.; Sonbhadra, S.K.; Agarwal, S. Monitoring COVID-19 social distancing with person detection and tracking via fine-tuned YOLO v3 and Deepsort techniques. *arXiv* 2020, arXiv:2005.01385.
- 43. Khosrawipour, V.; Lau, H.; Khosrawipour, T.; Kocbach, P.; Ichii, H.; Bania, J.; Mikolajczyk, A. Failure in initial stage containment of global COVID-19 epicenters. *J. Med. Virol.* 2020, *92*, 863–867. [CrossRef] [PubMed]
- 44. Desjardins, M.; Hohl, A.; Delmelle, E. Rapid surveillance of COVID-19 in the United States using a prospective space-time scan statistic: Detecting and evaluating emerging clusters. *Appl. Geogr.* **2020**, *118*, 102202. [CrossRef] [PubMed]
- Burke, R.M. Active monitoring of persons exposed to patients with confirmed COVID-19—United States, January–February 2020. MMWR Morb. Mortal. Wkly. Rep. 2020, 69, 245. [CrossRef]
- 46. Lienhart, R.; Maydt, J. An extended set of haar-like features for rapid object detection. In Proceedings of the International Conference on Image Processing, Rochester, NY, USA, 22–25 September 2002; Volume 1, p. I.

- 47. Viola, P.; Jones, M. Rapid object detection using a boosted cascade of simple features. In Proceedings of the 2001 IEEE Computer Society Conference on Computer Vision and Pattern Recognition, CVPR 2001, Kauai, HI, USA, 8–14 December 2001; Volume 1, p. I.
- 48. Compton, M.; Barnaghi, P.; Bermudez, L.; GarcíA-Castro, R.; Corcho, O.; Cox, S.; Graybeal, J.; Hauswirth, M.; Henson, C.; Herzog, A.; et al. The SSN ontology of the W3C semantic sensor network incubator group. *J. Web Semant.* **2012**, *17*, 25–32. [CrossRef]
- Yavari, A.; Jayaraman, P.P.; Georgakopoulos, D.; Nepal, S. ConTaaS: An approach to internet-scale contextualisation for developing efficient internet of things applications. In Proceedings of the Hawaii International Conference on System Sciences 2017 (HICSS-50), Hilton Waikoloa Village, HI, USA, 4–7 January 2017.
- 50. Bykov, S.; Geller, A.; Kliot, G.; Larus, J.R.; Pandya, R.; Thelin, J. Orleans: Cloud computing for everyone. In Proceedings of the 2nd ACM Symposium on Cloud Computing, Santa Cruz, CA, USA, 9 June 2011; pp. 1–14.
- Yavari, A.; Panah, A.S.; Georgakopoulos, D.; Jayaraman, P.P.; van Schyndel, R. Scalable role-based data disclosure control for the internet of things. In Proceedings of the 2017 IEEE 37th International Conference on Distributed Computing Systems (ICDCS), Atlanta, GA, USA, 5–8 June 2017; pp. 2226–2233.
- 52. Kua, J.; Armitage, G.; Branch, P. A survey of rate adaptation techniques for dynamic adaptive streaming over HTTP. *IEEE Commun. Surv. Tutor.* **2017**, *19*, 1842–1866. [CrossRef]
- Briscoe, B.; Brunstrom, A.; Petlund, A.; Hayes, D.; Ros, D.; Tsang, I.; Gjessing, S.; Fairhurst, G.; Griwodz, C.; Welzl, M. Reducing Internet Latency: A Survey of Techniques and Their Merits. *IEEE Commun. Surv. Tutor.* 2016, *18*, 2149–2196. [CrossRef]

**Disclaimer/Publisher's Note:** The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.